Author's response to reviews

Title: Video conferencing versus telephone calls for team work across hospitals: a qualitative study on simulated emergencies

Authors:

Stein R Bolle (stein.roald.bolle@telemed.no)
Frank Larsen (frank.larsen@telemed.no)
Oddvar Hagen (oddvar.hagen@telemed.no)
Mads Gilbert (mads.gilbert@unn.no)

Version: 5 Date: 20 November 2009

Author's response to reviews:

Tables have been reformatted. One table is kept as a table and now includes the information in rows and columns. Four tables are changed to appendixes, and are referred to in the text in square brackets. An appendix section appear before the Acknowledgements section.